Author's response to reviews

Title: Effects of epidural lidocaine analgesia on labor and delivery: a randomized, prospective, controlled trial

Authors:

Shahram - Nafisi - (nafisish@yahoo.com)

Version: 2 Date: 4 May 2006

Author's response to reviews: see over
The Title of my work has changed in this way:

EFFECTS OF EPIDURAL LIDOCAINE ANALGESIA ON LABOR AND DELIVERY: A RANDOMIZED, PROSPECTIVE, CONTROLLED TRIAL[ISRCTN32569437]